Author's response to reviews

Title: Social Networks, work and Network-Based Resources for the Management of Long Term Conditions: a framework and study protocol for developing self-care support

Authors:

Anne Rogers Professor (anne.rogers@manchester.ac.uk)
Ivaylo Vassilev Dr (ivaylo.vassilev@manchester.ac.uk)
Caroline Sanders Dr (caroline.sanders@manchester.ac.uk)
Susan Kirk Dr (susan.kirk@manchester.ac.uk)
Carolyn Chew-Graham Professor (carolyn.chew-graham@manchester.ac.uk)
Anne Kennedy Dr (anne.kennedy@manchester.ac.uk)
Joanne Protheroe Dr (joanne.protheroe@manchester.ac.uk)
Peter Bower Dr (peter.bower@manchester.ac.uk)
Christian Blickem Dr (christian.blickem@manchester.ac.uk)
David Reeves Dr (david.reeves@manchester.ac.uk)
Dharmi Kapadia Dr (dharmi.kapadia@manchester.ac.uk)
Helen Brooks Ms (helen.brooks@manchester.ac.uk)
Catherine Fulwood Dr (catherine.fulwood@manchester.ac.uk)

Version: 2 Date: 16 February 2011

Author's response to reviews:

Dear editor,

Thank you very much for your letter, please find below our response to your comments.

1) Please confirm, and if so explicitly state that this manuscript has provided an overview of the programme and that further published work will describe methods in detail.

We confirm that this manuscript has provided an overview of the programme and that further published work will describe methods in detail. This is now explicitly mentioned in the manuscript on p.15.

‘Further details describing the methods used in this research programme will be developed in separate publications.’

2) As this appears to be an overview of the programme, references to the 'project' needs to be removed from one of the section titles (and elsewhere).

p.12 the reference to the word ‘project’ is now removed.

3) On page 13, first sentence under 'Empirical study': "a randomized sample..." should read randomly sampled.
This is now changed to ‘randomly sampled’

We have also corrected some typos and would also like to ask you to add Dr. Catherine Fullwood as an author of the paper. All changes are made with track changes.

Please let me know if we may need to make any further changes to the paper.